Author's response to reviews

Title: Paying for treatments? Influences on negotiating clinical need and decision-making for dental implant treatment

Authors:

Catherine E Exley (catherine.exley@ncl.ac.uk)
Nikki S Rousseau (n.s.rousseau@ncl.ac.uk)
Jimmy Steele (jimmy.steele@ncl.ac.uk)
Tracy Finch (tracy.finch@ncl.ac.uk)
James Field (james.field@ncl.ac.uk)
Cam Donaldson (cam.donaldson@ncl.ac.uk)
J Mark Thomason (j.m.thomason@ncl.ac.uk)
Carl R May (c.r.may@ncl.ac.uk)
Janice S Ellis (j.s.ellis@ncl.ac.uk)

Version: 2 Date: 8 January 2009

Author's response to reviews:

Title checked
Author list checked
Country added
Abstract changed in format
Table - all gridline removed and all shading removed (I hope this is what you wanted it wasn't quite clear)